Author's response to reviews

Title: Comparative analysis of the human hepatic and adipose tissue transcriptomes during LPS-induced inflammation leads to the identification of differential biological pathways and candidate biomarkers

Authors:

Ewa Szalowska (ewa.szalowska@wur.nl)
Martijn Dijkstra (m.dijkstra@rug.nl)
Marieke G.L. Elferink (M.G.L.Elferink@rug.nl)
Desiree Weening (d.weening@med.umcg.nl)
Marcel de Vries (marcel.de.vries@med.umcg.nl)
Marcel Bruinenberg (m.bruinenberg@lifelines.umcg.nl)
Annemieke Hoek (a.hoek@og.umcg.nl)
Han Roelofsen (j.roelofsen@med.umcg.nl)
Geny M.M. Groothuis (g.m.m.groothuis@rug.nl)
Roel J. Vonk (r.j.vonk@med.umcg.nl)

Version: 2 Date: 2 December 2010

Author's response to reviews:

Dear Editor,

In this revised version of the manuscript we included a paragraph stating that our research protocols conformed the Helsinki Declaration and added information about the local legislation and patients' informed consent to participate in the study. Additionally we uploaded new versions of Figures 2, 3, 4,-5.

Kind regards,

Ewa Szalowska